Author’s response to reviews

**Title:** Socio-demographic determinants of the severity of locomotor disability among adults in Bangladesh: a cross-sectional study, December 2010-February 2011

**Authors:**

Ilias Mahmud (imahmudot@gmail.com)

Lynda Clarke (Lynda.clarke@lshtm.ac.uk)

George Ploubidis (g.ploubidis@ucl.ac.uk)

**Version:** 3  **Date:** 02 Jul 2017

**Author’s response to reviews:**

Editorial comments:

We thank the authors for the reply and changes made in manuscript. Please find some remaining editorial comments:

* Comment: revise the title of the figures and tables: e.g. title figure 1 appears twice, while the last figure is without title

Response: Corrected. Please see the revised version.

* Comment: add in the method section the range of the disability score (I expect: 0-220)

Response: we applied latent variable model to calculate disability score. Please see lines 147-50 (p 6-7)

We sincerely thank the reviewers and editors for their valuable comments and effort in improving our manuscript.